Panic Buying and Fake News in Urban vs. Rural England: A Case Study of Twitter During COVID-19

# Abstract

This paper explores the potential association between the spread of fake news and the panic buying behavior, in urban and rural UK, widely accessible on Twitter since COVID 19 was announced by the WHO as a global pandemic. It describes how consumer's behavior is affected by the content generated over social media and discuss various means to control such occurrence that results in an undesirable social change. The research methodology is based on extracting data from texts on the subject of panic buying and analysing both the total volume and the rate of fake news classification during COVID-19, through crowdsourcing techniques with text-mining and Natural Language Processing models. In this paper, we have extracted the main topics in different phases of the pandemic using term frequency strategies and word clouds as well as applied artificial intelligence in exploring the reliability behind online written text on Twitter. The findings of the research indicate an association between the pattern of panic buying behaviour and the spread of fake news among urban and rural UK. We have highlighted the magnitude of the undesired behaviour of panic buying and the spread of fake news in the rural UK in comparison with the urban UK.

# 1. Introduction

The emergence of the novel coronavirus (SARS-CoV-2) in December of 2019 has quickly led to a global pandemic claiming hundreds of thousands of deaths worldwide already (Roser et al., 2020). In the absence of an effective treatment or vaccine, researchers have pointed out that managing the pandemic response would require leveraging insights from the social and behavioral sciences, particularly with regard to non-pharmaceutical interventions and containing the spread of fake news and misinformation about COVID-19 (Depoux et al., 2020; Habersaat et al., 2020; Van Bavel et al., 2020).

In fact, the spread of misleading information about the virus has led the World Health Organization (WHO) to warn about an on-going "infodemic" or an overabundance of information— especially misinformation—during an epidemic (World Health Organization, 2020 in van dar Linken, et al. (2020); Zarocostas, 2020). This infodemic makes it harder for people to find trustworthy and reliable information when they need it.

Specifically for COVID-19, misinformation could have led to a higher number of cases and deaths, since people could have followed inadequate procedures to prevent contagion or treat the disease. The worldwide dashboard from WHO, as of October/2022, show more than 600 million confirmed cases and more than 6.5 million deaths due to COVID-19 (WHO, 2022). Since the number of cases and deaths could be affected by the spread of good or bad information, the study of consumer behavior during the COVID-19 pandemic, especially taking into account the dissemination of information through social media, may shed light to relevant insights related to technology advances and social challenges.

In this article, we ask four critical questions to help better inform societal response to the infodemic, namely; (1) what is the scope and reach of fake news about COVID-19

in the general population during COVID-19 period?, (2) what is the scope and reach of panic buying behaviour during COVID-19 period?, (3) what evidence is there to suggest that fake news about the virus is encouraging the public support for—and the adoption of— panic buying behavior?, and (4) how can insights from the digital divide be leveraged to effectively manage societal response to help limit the spread of influential fake news and limit the panic buying behavior?

These questions are relevant and timely issues for understanding better the social phenomenon related to panic buying. We use data from the United Kingdom, including posts on social media, to explore issues related to fake news, consumer behavior, and digital divide. More specifically, regarding the first question, fake news, especially during health crises may lead to inadequate sanitary behavior or treatment procedures that could cause higher number of deaths. From a societal perspective, the second question, panic buying can result in price surges and product shortage, making more vulnerable people to bear more severe consequences of the pandemic. The third question aims at investigating evidences of relationship between fake news and panic buying, advancing the implications of the first two questions. Finally, the fourth question aims at further investigating the main elements of the research, including the perspective of digital divide. In particular, we tackle the potential different effects of fake news in different public profiles in relation to access to technology, The remainder of the article is organized as follows. Section 2 discusses the extant literature

around the key themes of the paper: panic buying, fake news and digital divide. Section 3 presents the research methodology adopted in the paper, followed by the presentation of findings in Section 4. Section 5 discusses the association of fake news, panic buying during COVID 19 amongst twitter users, in context of the digital divide in the UK. Finally, Section 6 presents the conclusion, limitations, future research agenda, and implications for policy and practice.

#### 2. Literature review

#### 2.1 Panic Buying

# 2.1.1 Issues in panic buying

Crises and disasters proved to change consumption and consumer behavior (Forbes, 2017; Peck, 2006; Sheu and Kuo, 2020). This change could comprise undesirable incidences such as herd mentality, variations in investment decisions and in purchasing habits, particularly panic buying (Loxton et al., 2020).

Panic buying is the act consumers undertake through purchasing larger amounts of products than they used to buy when they perceive a disaster, product's shortage, or high price increase (Yoon et al. 2017; Yuen et al. 2020). Panic buying could lead to a shortage in necessities and medical supplies because this behavior can be widely spread (Thomas, 2014), which in turn leads to additional unavailability of resources that can exacerbate the impact, creating further panic buying (Hall et al., 2020). A serious consequence of such behavior lies in preventing vulnerable groups to acquire their basic needs (Besson, 2020). Previous research showed contradictory findings concerning the role of consumers in increasing the unavailability of goods due to the higher demand (Tsao et al., 2019; Quarantelli, 1999).

Panic buying was clearly witnessed during the early stages of COVID-19 when consumers started to stockpile medical supplies, groceries, hand sanitizers, and even toilet paper (Barr, 2020; Collinson, 2020) implying empty shelves in many stores (Cogley, 2020; Mao, 2020). These changes in consumption are due to the "interplay between shifts in both consumer demand and availability of supply" (Hall et al., 2020), and the extent of this consumption

displacement varies through the stages of a disaster (preparedness, response, recovery, postevent planning and mitigation) (Anderson et al., 2020).

Moreover, Non-Pharmaceutical Interventions (NPIs) adopted to face COVID-19 such as limiting people's mobility and personal contact have exacerbated the problem through affecting consumption displacement occurrence over four dimensions: (i) where (spatial aspect): limiting the movement of the consumers; (ii) when (temporal aspect): time variation for enabling/disabling access to different merchandises; (iii) what and why: types of products demanded based on psychosocial influence; and (iv) how: changing the way and process of purchasing (Hall et al.,2020).

The scientific literature identifies a number of factors that lead to and fuel panic buying. First, when consumers perceive a future shortage in certain products that would affect their lifestyle, they may anticipate the purchase to avoid regret not possessing it in the future (Yoon et al., 2017; Yuen et al., 2020). Second, due to the risk perceived by consumers during times of crises and uncertainty of their durations, consumers tend to make irrational decisions and change their buying patterns (Slovic et al., 2004; Sim et al., 2020). A third reason for panic buying is attributed to the loss of control consumers feel, which drives them to acquire more products to regain a sense of control and stress alleviation (Ballantine, 2013). Furthermore, consumers can be influenced by others' opinions and behaviors in their society (Yuen et al., 2020). Another factor that drives panic buying behavior lies in the encouragement of some governments to prompt citizens to accumulate goods to avoid their reliance of government's aid (Kulemeka, 2010). In addition, the fear of a complete store shutdown and lockdown posed by governments shortens the operating hours of stores and limits the shopping time and location for citizens (Islam et al., 2020), making consumers more willing to stockpile products. In a nutshell, the reasons behind panic buying are all a consequence of a change in behavior of the individuals, communities, and governments when a disaster occurs.

#### 2.1.2 Understanding Consumer's panic buying behavior

Various studies attempted to depict the consumer's decision-making process that leads to panic buying (e.g., Campbell et al, 2020, Prentice et al, 2022, Sing et al, 2021, Huan et al, 2021). The study from Billore and Anisimova (2021) provides an extensive literature review

on panic buying and the paper from Cruz-Cárdenas (2021) performs an analysis of the literature on COVID-19, consumer behavior and society. Particularly, taking into account the changes in society, especially due to technological advances, the content generated on social media provides a rich source for researchers to analyze the behavior of consumers during crises. Hence, the extant literature offers many frameworks developed to examine how consumers' spending behavior increases during times of crises.

A thorough review of the relevant literature revealed that the Theory of Planned Behavior (TPB) developed by Fishbein and Ajzen (2011) is widely referred to in understanding pandemic-driven consumer behavior (Wang et al., 2022). For example, through gathering both quantitative and open questions from consumers in Germany, Lehberger et al. (2022) concluded that stockpiling behavior is due to three main drivers: attitude, subjective norms, and fear of future unavailability. Furthermore, through an online survey on 371 respondents from Malaysia, Tan et al. (2021) proved that attitude and subjective norms have an impact on panic buying unlike the third construct in TBP: perceived behavior control that was not found a predictor of panic buying. Moreover, the findings of Tan et al. (2021) showed that online news influence attitude, subjective norms, and Perceived Likelihood of being Affected (PLA). They noted also that PLA affects subjective norms as individuals tend to be attributed to close groups when perceiving a risk.

In addition, Laato et al. (2020) proposed a model based on the Stimulus Organism-Response (SOR) framework developed by Arora (1982) to examine the relations among: online information source exposure, information overload, perceived severity, cyberchondria, self-isolation intention, intention to make unusual purchases, self-isolation, self-efficacy, and purchasing self-efficacy. The authorse tested their research model through conducting an online survey with 211 Finnish respondents. Their findings show a strong link between self-isolation intention and intention to make unusual purchases. In addition, exposure to online information sources has an impact on cyberchondria and on information overload, which in its turn affects cyberchondria. Moreover, both intentions to make unusual purchases and to online information sources has an impact on cyberchondria and on information overload, which in its turn affects cyberchondria. Moreover, both intentions to make unusual purchases and to online information sources has an impact on cyberchondria and on information overload, which in its turn affects cyberchondria. Moreover, both intentions to make unusual purchases and to online information sources has an impact on cyberchondria and on information overload, which in its turn affects cyberchondria. Moreover, both intentions to make unusual purchases

and to self-isolate are a result of perceived severity and cyberchondria. The SOR framework was also referred by Islam et al. (2020) in combination with the Competitive Arousal model to build a framework in which they used the same online survey method but on respondents from the US, China, India, and Pakistan. They concluded that external stimuli, such as limited quantity and time scarcities, fuel consumers' arousal, which drives them to impulsive and obsessive buying. Furthermore, Li et al. (2021) extended the SOR model and combined it with the dual-system theory to understand how consumers are driven into panic buying. Their analysis of data collected from 508 residents in Singapore showed that impulsive buying decision results from reflective thinking and environmental stimuli (such as, perceived susceptibility and severity of the crisis, social norm, and social influence). Such environmental stimuli can affect consumers' perceptions of scarcity and affective response (panic and fear of products' unavailability).

Panic, stress, and anxiety were also highlighted by several research such as, Sneath et al. (2008) and Barnes et al. (2020). Based on life event theory, Sneath et al. (2008) developed a model and tested it through collecting data from 427 US Gulf Coast residents who were affected by Hurricane Katrina. Their findings showed that both perceived lack of control and loss of possession lead to stress. Accordingly, stress results in a depressive state. Being depressed, people's behavior tends towards compulsive and impulsive buying. It was proven also that age, gender, income, and insurance coverage do not moderate the relationship between depression and impulsive buying. They stated that when a disaster occurs, impulsive buying prevails as a rational behavior rather than compulsive buying. It is worth mentioning that impulse buying is not always associated with emotions as there exists sometimes a cognitive aspect that moderates the relationship between impulse buying trait and consumers' buying behaviors (Rook and Fisher, 1995). Barnes et al. (2020) through the use of the Compensatory Control Theory (CCT) have conducted a study to prove that anxiety and fear feelings negatively affect consumers' perceived control, which in its turn increases purchasing behavior especially in case of utilitarian's low quality. In their study, they relied on big data set gathered from Twitter users in Italy. Over and above general depression, stress, and anxiety (negative affect), Gallagher et al. (2017) highlighted the specific effect of Anxiety Sensitivity (AS) on compulsive buying. Their study on a sample of 437 Canadian undergraduates concluded that two of the three dimensions of AS: Cognitive concerns (e.g.,

fear of losing control) and Physical concerns (e.g., fear of having a heart attack) have an impact on the tendency to buy compulsively. Interestingly, the third dimension of AS, Social concerns (e.g., fear of public ridicule) was not proved pivotal in compulsive buying behavior.

Governments' planning and intervention during crises and their influence on consumer behavior were also highlighted by Prentice et al. (2022), Loxton et al. (2020), Nakano et al. (2021), etc. They concluded that panic buying was to a large extent a reaction to the legislations and restrictions set by governments. Based on the scarcity principle, crowd psychology and contagion theory, Prentice et al. (2020) gathered data from 341 consumers from the USA and Australia who experienced panic buying, and concluded that public health risk mitigation measures enforced by governments to control the virus spread (such as, lockdowns and social distancing) were perceived as a sign for resource scarcity. In addition to the influence of the government, Loxton et al. (2020) employed the volume and timing of consumer spending patterns of American and Australian markets and proved that the behavior of consumers during COVID-19 is the same during other crises in terms of panic buying, herd mentality, and consumption's priority (based on Maslow's Hierarchy of Needs Model). However, others see that COVID-19 amplified panic buying behavior due to internalization, isolation measures, absence of vaccine for a long time and uncertainty about duration of the pandemic and the extent of effectiveness of the vaccine (li et al., 2021; Loxton et al., 2020; Dulam et al., 2021). Nakano et al. (2021) proved a strong association between the timing of two government policy settings and two corresponding waves of consumers' panic buying of 173 product categories in Japan. In addition, they embraced a segmentation approach to understand the difference in behavior among different consumers in terms of demographics and psychographics. Their study relied on data collected through 3 different sources: extracted purchase data, device log data from television and mobile devices, and survey feedback from 968 respondents.

The work of Nakano et al. (2021) pinpointed key critical aspects: the time factor and the demographic, psychology, and personal traits of consumers. First, the influence of the stage of both the pandemic and corresponding reaction of the government on the hoarding approach of the consumers, such factor is confirmed by the studies of Hall et al. (2020) and Dulam et al. (2021) that state that the over purchasing decision making process during crises

is phase dependent. As for the difference in buyers' personal characteristics, Bentall et al. (2021) developed a psychological model of over-purchasing derived by the animal foraging theory that combines the different variables that lead to over-purchasing. Based on data collected from online surveys on 2025 respondents from the UK and 1041 ones from Ireland. Their findings proved that panic buying is affected by household income, presence of children at home, psychological distress (depression, death anxiety), mistrust of others (paranoia), and ability to reflect about reassuring messages. Interestingly, their results revealed that overpurchasing occurred in Ireland more than the UK and in Urban more than rural areas. Individual differences were also researched by Dulam et al. (2021) through the use of the agent model to examine how the mental, emotional, behavioral states of the consumers are important predictors of their purchasing decision over a 6-stage decision-making process: need recognition, information search and processing, factor valuation, decision, purchase, and purchase evaluation. They developed a simulation tool and tested it through conducting a questionnaire survey on 2000 households following the distribution and count of demographic data in Japan. Furthermore, the simulation tool created by Dulam et al. (2021) studied consumer behavior in accordance with the supply chain, as smoothing the supply chain process would primarily have a positive outcome on reducing panic buying (e.g., Lee et al. (1997) and Sterman and Dogan (2015). Their conclusion stressed on the necessity of an effective policy setting to mitigate the impact of over-purchasing on the supply chain, and of situation-dependent rationing measures. Understanding consumer behavior during crises and its impact on the supply chain was also investigated by Zheng et al. (2020) who emphasized on consumer's social learning behavior to anticipate supply disruption risk. Therefore, retailers could optimize their inventory strategies based on the extent of social learning that a second group of consumers acquire from the initial panic intensity of a first group.

It is important to highlight that this social learning effect can be aggravated by social media, which is a major player in influencing consumer behavior. Social media platforms have been impacting society costumes and also buying patterns. Particularly, as suggested by Appel et al. (2020), Specifically, during the Covid-19 pandemic, the guidelines for social distancing and the lockdown enforcements implied more information interexchange through social media Naeem and Ozuem (2021a).

Nowadays, social learning is evidently more intense due to the existence of social media. Social media enabled the propagation of news and ideas to wider groups -as the number of Internet users is increasing continuously (Arafat et al., 2020)- who are influenced by the views and perceptions of their related friends and acquaintances. As such, social media had a prominent part in fueling consumers' buying panic intention (Appel et al., 2020). Lockdown measures and social distancing resulted in more engagement and socialization over social media (Muqadas et al., 2017; Naeem and Ozuem, 2021a). Undoubtedly, social media has connected people together during the pandemic, facilitated information sharing, and has exerted a social change nurturing a "constructive voice behavior" (Bhatti et al., 2020) through online ratings, social motivation, influencers, etc. (Alalwan, 2018); Hence, shifting from individual to collective consumers' purchasing activities (Thomas et al., 2020), and assisting consumers in making better buying decisions (Alalwan, 2018; Baker Qureshi et al., 2019). On the other hand, the social change carried out by social media has sometimes led to negative consequences. The ease of generating and propagating content resulted in an uncontrolled immense spread of fake news. Since the start of the pandemic, social media became a rich medium for broadcasting plenty of rumors, misinformation, and news' magnification because people are usually attracted to shocking content (Cogley, 2020; Hou et al., 2020; Mao, 2020). Therefore, social media has increased fear and anxiousness, which has driven consumers to change their buying patterns and to stock-pile different utilitarian products (Naeem, 2021).

#### 2.2. Fake News

In the information age, the volume of data made available from official and non-official entities grows more and more during the pandemic. Rather than increasing transparency and assisting citizens and governments in decision-making, increasingly more accessible information about COVID-19 – yet contradictory - seems to confuse even more people, delaying the improvement of recovery whether in health or economics terms. Avoiding the spread of misinformation and fake news in the first place could therefore shunt the reluctance to follow governmental guidelines. Misinformation and fake news can amplify humanity's greatest challenges. A salient recent example of this is the COVID-19 pandemic, which has bred a multitude of falsehoods even as truth has increasingly become a matter of life-and-

death; people share false claims about COVID-19, in part, because they simply fail to think sufficiently about whether or not this content is accurate when deciding what to share (Pennycook, et al., 2020).

Users nowadays receive information mainly through the internet, especially during the COVID-19 when most people have to stay at home. These users receive such information through an ever more advanced algorithm of public and content selection, regardless of the media they choose. This process is known for direct or indirect creating the so-called "information bubbles", in which users receive only, in a consistent manner, one type of information. This variation can go from political opinions to social engineering desires.

One consequence of this during Covid-19 is that users can be sometimes confused with the amount of information being spread, especially when it seems to be sharply divergent from media to media with potential bias. The information during Covid-19 is worldwide being broadcasted in an decentralised means, and fake news is increasingly more difficult to set apart from trusted news. Traditional media, such as television, printed newspapers and radio, continue to play a significant role. However, user-generated content and even official sources on social media are having a more significant share now than ever before. This decentralised content generation can be referred to as crowdsourcing and has a vital role in understanding social behaviour (Chamberlain et al, 2021). Decentralised content generation is not only a phenomenon that has an impact on better understanding the data, but also a necessary mean of information sharing. Government and private sector heavily depend on the use of the social media platforms for rapid information sharing, and during COVID-19, this dependency has only increased (Chan et al, 2020). Analysing such contents during the pandemic has proven to be effective (Lopez, Vasu and Gallemore, 2020) for understanding users' trend of social media. Not surprisingly, rumours and misinformation are a topic of great interest in understanding such behaviour during the pandemic (Tasnim, Hossain and Mazumder 2020).

It is unclear exactly how people start endorsing fake news, but their high level of circulation online is a recognized contributing factor, as seeing an information makes people more likely to believe in it, and seeing it several times re-enforce the view that it could be true, a psychological phenomenon called the illusory truth effect, (Roggeveen and Johar, 2002; Unkelbach, Koch, Silva, & Garcia-Marques, 2019).

Being able to avoid fake news during the pandemic could not only help the individual receive the best information for their health by taking the best choice possible but could also make governmental guidelines clearer and easy to follow, hence more effective.

The WHO has classified the information overload about COVID-19 as an "infodemic". The infodemic has become such an important subject with some effort made to help people classify information in categories as information; misinformation; disinformation and malinformation (Baines, D. & Elliott, R. J., 2020). This undesired information is formed by fake news, rumours, and conspiracy theories, promotion of fake cures, panic, racism, xenophobia, and mistrust in the authorities, among others (Alam, et al., 2020).

Some evidence show that it is possible to analyse user behaviour by analysing tweets. A research project, for instance, has analysed 1000 posts and concluded that although false information was more frequent than legit ones, these received less retweets, indicating users could to some degree use this kind of information to filter news (Pulido, Cristina M., et al.,2020).

Other researchers analysed other social medias and found that the level of misinformation varies according to the social media platforms, having pointed that twitter is among the biggest social networks the one with the most questionable sources (Cinelli, M, et al., 2020).

There is, still, demand for analysis that can go beyond the simple true and fake classification, which could perhaps translate some deeper political nuances and use information beyond text to allow a better automated analysis of content being produced online during the pandemic (Knight, J., et al., 2020).

There are examples of successful uses of Twitter to forecast off-line behaviour, for example, film box office gross (Asur and Huberman, 2010; Hennig-Thurau, Wiertz and Feldhaus, 2015), book sales (Gruhl, et al., 2005), and music sales (Frick, Tsekauras, and Li, 2014). Therefore,

Twitter data is an appropriate context to find the association between fake news trend and panic buying off-line behaviour.

The digital divide has been a major focus of online research; (Blank, 2016). Digital inequality can take many forms that have been explored in the United Kingdom as well as elsewhere, (Blank, 2016). In Britain, the Oxford Internet Survey (OxIS) has charted 10 years of trends in the online population (Dutton & Blank, 2011, 2013). These studies document that British Internet users have been younger, better educated, and wealthier than the off-line population since the earliest wave in 2003. Some differences between the online and off-line populations have disappeared, such as the gender gap which was important in the early 2000s but disappeared by 2011. Students have been the most likely to use the Internet, although employed people have been closing the gap. Retired people are least likely to be Internet users. Disabled people are about half as likely to use the Internet as nondisabled, although this gap has been declining. Black and Asian minorities are more likely to use the Internet than Whites. Urban–rural differences are not significant, ;(Blank, 2016). We further discuss the recent developments regarding digital divide in the UK and consider its potential impact in the context of fake news and off line behaviour for the digitally excluded population in the UK.

# 2.3 Digital Divide

Digital divide refers to "inequalities in access to internet, extent of use, knowledge of search strategies, quality of technical connections and social support, ability to evaluate the quality of information and diversity" (DiMaggio, 2010, p 310). Rural communities are normally located remotely from main cities and hence are in greater need of better digital connectiveness to facilitate a better of quality of life, yet this is not always the case (Salemink et al., 2017). A common finding in the extant research conducted on digital exclusion and inequalities in the Western countries show that rural telecommunication infrastructure is inferior to urban areas (Philip et. Al., 2017; Gerli and Whalley, 2021). Given its profound impact on population's quality of life, digital divide has been investigated extensively over the last two decades in the UK and indeed occupied a considerable policy space (Serafino, ,2019). Despite that, global media had a tendency to portrait digital connectivity, especially in the

West, as pervasive and abundant, this is not always the case, for example in the UK the ONS (2019) figures have shown that despite a significant improvement in the number of adults who have either never used the internet or have not used it in the last three month, still there is 5.3 million adult who falls within this category. This is a worrying figure as this means that this population could miss out on a number of benefits from being connected to the internet such as earning potential, employability benefits, retail transaction benefits, communication benefits and time saving in accessing government and banking services (Serafino, 2019). This area of concern has been targeted by government policy on two dimensions: the socioeconomic and technological infrastructure (Sparks, 2013). Where the earlier is addressed through enhancing stakeholders' ICT skills, digital literacy i.e especially in rural areas, the latter is addressed by further investing in psychical infrastructure supporting connectivity to the internet (Blank and Groselj, 2015; Gerli and Whalley, 2021). Here, policymakers in the UK have designed a number of initiatives to address these forms of digital divides for example: The UK Gigabit Programme or 'Project Gigabit' which is a £5 billion government infrastructure project focusing on delivering a fast and reliable digital connectivity for the entire country (Department for Digital, C., Media & Sport, 2021) Notably this massive investment was also underpinned by the government's 'Barrier Busting Task force' which was initiated in 2017 to address any barriers preventing "the fast, efficient and cost-effective deployment of gigabitcapable broadband and improved mobile coverage, including next generation 5G technology." (Department for Digital, C., Media & Sport, 2021). Previous initiatives focusing solely on rural areas were also introduced in 2017: Rural Gigabit Connectivity Programme. Yet the digital challenges to rural population remain.

Despite the above-mentioned efforts to address the digital divide, one would argue that these inequalities have manifested since the onset of the COVID-19 pandemic, as more than half of the UK participants surveyed by the ONS in 2018 has shown that they use the internet to look for health related information (Watts, 2020). As such and even if this figure has improved by March 2020, it means that a significant part of the population must have suffered the bitter consequences of the digital divide. Moreover, during lockdowns the digital divide has reportedly negatively affected mental health and general wellbeing of those digitally excluded (Watts, 2020). In the same vein, young adults who have grown up in a relatively digitally driven environment, where using the internet is not a matter of preference but rather a lifestyle and a platform for communication that has taken-over conventional methods of

communication (making landline calls or even mobile phone calls). Whereby mobile phones are mainly used as smart devices to communicate via social media. This group who are living in rural areas in the UK face digital connectivity difficulties in the form of a relatively poor internet services which excludes these group from participating in online activities as their fellow urban dwellers (Philips et. al., 2017). Other groups are also impacted as the ONS survey in 2018 shows that 60% of the internet users use it for social networking (Serafino, 2019).

Furthermore, the impact is not only limited to health and social communication, but also, to commercial transactions. In the ONS 2018 survey, 65% of the internet users indicated that they use it to find information about goods (Serafino, 2019). In fact, commercial transactions and use of social networking platforms became interrelated, with the rise of e-word of mouth (e-WOM) where users' opinions such as friends, peers and family regarding products or services can influence other users' decision of buying (Castellano and Dutot, 2017). A negative side though to this digital engagement can also lead to viral spread of misinformation (Naeem, 2021). However, taking into consideration the digital divide along with non-internet users in rural areas, one could reasonably argue that the impact of e-WOM as well as spread of misinformation could be less prevalent due to digital exclusion and hence this population could potentially be protected from the adverse effects of participation in social networking platforms such as "fake news, hatred and creating racism during epidemics and civil unrest" (Kadam and Atre, 2020). Having said that, with the progression and successful rolling out of high-speed internet and programmes to enhance the digital skills of rural dwellers, this picture could change and along with reaping the benefits of being digitally connected, unintended consequences can include being exposed to the adverse effects noted above. However, taking into consideration that in some cases the digital skills courses could be too far from rural areas, expensive or of a questionable quality (Parliament.UK,2019) could be translated to a persistent users' problems that can widen the exposure to these adverse effects of digital connectivity.

Of note, digital divide observed between urban and rural areas can also be reflected on social media usage and which platform to use, as these decisions can be socially constructed (Hine, 2020). In fact, digital divide on Twitter can be slightly different from the traditional notion of digital divide (Blank, 2015). More specifically, research has shown that Twitter users are young, well-educated, single and wealthy elites (Blank, 2015). However, while this profile is useful to know as it gives insights to engagement of these segments with Twitter, Blank (2015)

warns that "twitter data is not proxy for research on the population as a whole or even the subset of the population that is online. Instead, Twitter users reflect the interests, values, skills, priorities, and biases of elites" (Blank, 2015, p.13). Reasons for not using Twitter are more likely socio-cultural rather than technical/financial (Hine, 2020). Moreover, Twitter users are 12-38 % more likely to participate in an activity than non-Twitter users. This shows that a prudent interpretation is needed when Twitter data is analysed and especially in terms of generalising findings based on Twitter data to the general population of internet users or offline individuals. For example, using Twitter data to forecast election results would lead to inaccurate or possibly wrong conclusions (Huberly, 2015). Fortunately, there are some exceptions to using Twitter to forecast off-line behaviour and as such generalise to the population to which these users are part of, such as in the case of commercial products (Blank, 2015).

Against the above background, this paper attempts to investigate the scope and reach of fake news and the association with panic buying behaviour in the context of an existing digital divide in the UK.

## 3. Research Methodology

#### 3.1 Model and data

The data analysis step in this paper took place in two stages. Initially, uncharacterized posts were collected from the social network *Twitter* from England. The collection, carried out through the use of API, collected data between March 2020 and October 2021 and used posts related to *panic buying* as search criteria. This search method uses *Twitter*'s mechanism of search for returning the associated posts that meet the search criteria.

The search was performed at the following locations from a manually entered list of 51 Towns: Bath, Birmingham, Bradford, Brighton and Hove, Bristol, Cambridge, Canterbury, Carlisle, Chelmsford, Chester, Chichester, Coventry, Derby, Durham, Ely, Exeter, Gloucester, Hereford, Kingston upon Hull, Lancaster, Leeds, Leicester, Lichfield, Lincoln, Liverpool, London, Manchester, Newcastle upon Tyne, Norwich, Nottingham, Oxford, Peterborough, Plymouth, Portsmouth, Preston, Ripon, Salford, Salisbury, Sheffield, Southampton, St Albans, Stoke-onTrent, Sunderland, Truro, Wakefield, Wells, Westminster, Winchester, Wolverhampton, Worcester, York.

Because the search depends on the location chosen by the user, the search was also performed by Counties (totalling 34 counties) to ensure a larger number of posts, since the search selection criteria restrict our results, hence why we search for more posts. The list used consisted of: *Somerset, West Midlands, West Yorkshire, East Sussex, Bristol, Cambridgeshire, Kent, Cumbria, Essex, Cheshire, West Sussex, Derbyshire, Durham, Devon, Gloucestershire, Hereford and Worcester, East Riding of Yorkshire, Lancashire, Leicestershire, Staffordshire, Lincolnshire, Merseyside, London, Greater Manchester, Tyne and Wear, Norfolk, Nottinghamshire, Oxfordshire, Hampshire, North Yorkshire, Wiltshire, South Yorkshire, Hertfordshire, Cornwall.* 

This location was used to cross-reference the data with the official classification database for assessing the level of rurality of the results obtained: *Rural Urban Classification (2011) of Counties in England*<sup>1</sup>. The classification, according to the source, is based on a numerical scale ranging from 2 to 6, with 2 being Predominantly Rural and 6 being Predominantly Urbanized. No county is classified, according to the source, as 1.

The total number of posts collected in the specified period was 1602 posts. Because it is not a value considered expressive, the distribution along the rurality classifications was not balanced and for this reason the group with 2-3 classifications were analyzed as being rural and 4-5-6 as being urban classifications. Possible implications of this scenario are discussed on the implications and future research section.

For further classification as to the veracity of the information (*fake news* classification) we used the textual database *LIAR Dataset*<sup>2</sup> (Wang, 2017). The LIAR Dataset consists of small statements manually classified for truthfulness, which range from pants-fire (most false option) to true, and deal with different topics. Other information in the dataset includes the authorship (speaker), context, state, political party (when applicable) and others. The ratings were then aggregated into two main groups, namely True statements and False statements.

<sup>&</sup>lt;sup>1</sup> Source: Office for National Statistics licensed under the Open Government Licence v.3.0

<sup>&</sup>lt;sup>2</sup> " liar, liar pants on fire": A new benchmark dataset for fake news detection.

This procedure aims at classifying collected panic buying twitter-posts in a binary fashion. An example of the aggregation process is shown in table XX.

Original Label	<mark>Used Label</mark>	Statement
True	True	Building a wall on the U.SMexico border will take
		literally years.
<mark>Mostly-true</mark>	<mark>True</mark>	Marijuana is less toxic than alcohol.
Half-true	True	Obamacare was the Republican plan in the early 90s.
<mark>Barely- true</mark>	False	Hillary Clinton said gun confiscation would be worth considering.
False	False	Half of children struck by cars near schools are hit by parents driving children to school.
Pants-fire	<b>False</b>	Evidence shows Zika virus turns fetus brains to
		liquid.
Despite the simplif	ication process in fo	prcing binary classification, the results obtained by the
authors of the data	base (Original Label	), when training a supervised model was approximately
<mark>0.27</mark> (Wang, 2017)	possibly evidencin	g that there is an excessive subdivision of the original
<mark>data. Thus, it is exp</mark>	ected that the resul	ts of this step will only guide the analysis of the volume
of data collected	on panic buying, h	ence it is beneficial to have a more comprehensive
classified sample.		

We then used the model for classification on the data collected from *Twitter* regarding panic buying, from the first step. Results are presented on the end of this section comparing the volume of posts containing panic buying issues with the results of this veracity classification, both in total values and aggregated by urbanization groups 2-3 (rural) vs 4-5-6 (urban).

# 3.2 Natural language processing overview

Before classifying the text as *fake news*, we need to adapt both the text used in the training (coming from the *LIAR Dataset*<sup>2</sup>) and the collected posts themselves in order to classify them later. This process of textual analysis is commonly called *Natural Language Processing* (NLP). The following is a summary of the main points adopted in this step.

In this work, the representation of words in the *bag-of-words*<sup>3</sup> (Zhang et al., 2010) format was adopted. This representation is the transformation of sentences into numeric vectors that

<sup>&</sup>lt;sup>3</sup> Fixed-length representation of individual tokens, disregarding the relationship with neighboring tokens and order.

enable the computer to perform the mathematical operations of classification. The first step is the separation of words into tokens<sup>4</sup> and the application of preprocessing (cleaning), followed by numerical conversion into vectors and finally the relative calculation of the frequency of each term in the vocabular (term frequency-inverse document frequency or tfidf). The following is an example of this for two sentences: Today I read a Panic Buying news and Today winter starts:

The phrases

		Т	oday I	read	a new	s art	icle a	bout	Panio	c Buyi	ng				
and															
					Today	ı beg	ins w	inter							
beco	me respec	tively													
	today		1	I rea		nd news		article		about pa		nic buy		ving	
and															<u> </u>
			today b		beg	gins win		iter	-						
in the	standardi	zedtoke	o conv	arsion	The	conv	ersio	ninto	vect	orso	f tho fi	rct c	onton	re fo	llows
mune	stanuarui			213101	I. THE	COIIV	ersio	minu	veci	0150	i the h	1515	enten		10005
oday	Ι	read	ne	WS	's artic		about		panic		buying		begins		winte
1	1	1	-	1	1		1	!	1		1		0		0
and f	or the seco	ond sente	ence												
oday	Ι	read	ne	ws	article		about		panic		buy	ing	begi	ns	winte
1	0	0	(	)	0		0	0		)	0		1		1
Anot	her import	ant step	is the d	correc	ction b	y the	e inve	erse fi	reque	ency c	of tern	ns in	sente	nces	. This
mear	ns that we	penalize	words	that	appea	r wit	th hig	h freo	quen	cy in s	severa	al po	sts. Tł	nis fo	ollows
for b	oth senten	ices, resp	ectivel	у											
oday	Ι	read	ne	WS	arti	cle	abo	out	ра	nic	buy	ing	begi	ns	winte
1/2	1/1	1/1	1,	/1	1/.	1	1/	′1	1,	/1	1/	1	0/1		0/1
and		1			1		1				1		1		1
odav	1	read	ne	ws	arti	cle	abo	out	ра	nic	buy	ing	begi	ns	winte

<sup>4</sup> Individual part of a sentence having semantic value.

1/2	0/1	0/1	0/1	0/1	0/1	0/1	0/1	1/1	1/1	

In this way we can work with a large volume of data in an optimized way such that allows us to analyze the relationship of each word in the same vector space with the entire vocabulary. Other data cleaning processes involve eliminating *stopwords*<sup>5</sup>, limiting the vocabulary to words with a frequency of appearance of less than 70% of the documents, and maintaining only alphanumeric words with at least two characters.

# 3.3. Machine learning

The model adopted in the training of the *fake news* classification model is relatively simple (especially regarding *deep neural networks*), but of great effectiveness for classification activities, being a frequent choice for *NLP* text analysis (Kim et al., 2006)(McCallum and Nigam, 1998). The model, called *Multinomial Naive Bayes* and belonging to the class of supervised models, is said to be naive because it considers that all observation pairs are independent for a given class *y* (in this case, corresponding to being false or not).

In text analysis, this could mean not analyzing the relationship between neighboring words, which would be a detriment in context analysis. However, this does not usually happen due to the adoption of *n*-grams<sup>6</sup> and, mainly, by statistical approximation of the likelihood function for a high number of tokens in the corpus, as shown below for the dependent variables (our *n* tokens, or *n* words)  $x_1$  through  $x_n$ :

$$P(y|x_1,...,x_n) = \frac{P(y)P(x_1,...,x_n|y)}{P(x_1,...,x_n)}$$
(1)

Using the naive conditional independence assumption

$$P(x_i|y, x_1, \dots, x_{i-1}, x_{i+1}, \dots, x_n) = P(x_i|y),$$
(2)

for all *i*, this relationship is simplified to

$$P(y|x_1, \dots, x_n) = \frac{P(y) \prod_{i=1}^n P(x_i|y)}{P(x_1, \dots, x_n)}$$
(3)

Since  $P(x_1, ..., x_n)$  is constant given the input (it depends on the corpus used for training, and not on the single sentences to be classified), we can use the following classification rule:

<sup>&</sup>lt;sup>5</sup> words or terms irrelevant to the context

<sup>&</sup>lt;sup>6</sup> Combination of *n* words into a single *token*, e.g. *social\_media*.

$$\hat{y} = \arg\max_{y} P(y) \prod_{i=1}^{n} P(x_i|y),$$
(5)

And this way we can know which class has the highest probability of appearing, considering the individual contributions of each word relative to the class.

To implement the *MultinomialNB*<sup>4</sup> model, the *Scikit-learn*<sup>7</sup> package in the Python programming language was used, using the *tf-idf* inverse frequency count correction (Zhang, 2004). This model is similar to the one presented previously, parameterized by the vectors  $\theta_y = (\theta_{y1}, ..., \theta_{yn})$  where  $\theta_{yi}$  is the probability  $P(x_i | y)$ , with the difference of adding a smoothing parameter  $\alpha$  in order not to bias the model with words that are infrequent (= 0) in the test phase but may appear in the classification phase.

$$\hat{\theta}_{yi} = \frac{N_{yi} + \alpha}{N_y + \alpha n} \tag{6}$$

The parameter choices for the function are as presented in the *Parameter Tunning* subsection.

#### 3.4 Parameter tunning

The grid search method was used to search for the optimal combination of parameters to maximize classification accuracy. Two normalization methods were tested in the *df-idf* calculation: *L1-regularization*<sup>8</sup> vs *L2-regularization*<sup>9</sup>. Finally, for the  $\alpha$  parameter *Laplace* smoothers ( $\alpha = 1$ ) and *Lidstone smoothers* ( $\alpha = 10^{-1}$ ,  $\alpha = 10^{-2}$ ,  $\alpha = 10^{-3}$  and  $\alpha = 10^{-4}$ ) were analyzed. All results used factor 3 *cross-validation*, totalling 10 candidates and 30 fits. The criteria for accuracy was measured with *F1-Score*<sup>10</sup>.

With an average accuracy of 86% for the fake news training dataset, the optimal parameters chosen to classify the panic buying posts were  $\alpha = 10^{-2}$  and *df-idf* with *L2-regularization*.

#### 4. Findings and Results

After applying the previous model to the panic buying posts, we were able to compare how many of the collected posts could be classified as fake news according to this model. On

<sup>&</sup>lt;sup>7</sup> https://scikit-learn.org/stable/modules/naive\_bayes.html#multinomial-naive-bayes

<sup>&</sup>lt;sup>8</sup> L1 is just the sum of the weights

<sup>&</sup>lt;sup>9</sup> L2 is the sum of the square of the weights

<sup>&</sup>lt;sup>10</sup> Harmonic mean of the model's precision and recall

average, 33% (535) of the collected posts were classified as *fake news*. As the collected posts were collected with the information of the location, we can cross-reference this data with the rural and urban areas classification and analyse how both groups behave. The results for urban areas are presented on figure X.



Figure 1 - Word cloud - Urban, September 2021

Words like fuel, shortage, petrol and driver show a strong relationship to the origin of this

peak, indicating that they originated from a connected sharing pattern. Quantitative results

for the rural cases are presented in figure X.



Unlike the urban case, the rural scenario shows 41% of its posts in the first half of the analysis, and 59% in the second half of the analysis. Two high-intensity periods of posts occurred in mid-March 2020 and September 2021, with total percentage of rural posts at 33% and 49% respectively, with other posts distributed over the remaining months. The relative high volume of collected posts only coincides between rural and urban posts in the month of September 2021. It can be seen that when evaluating fake news about panic buying, these results appear to coincide with relevant waves of cases of covid-19 which occurred in the UK: in March to May 2020 and in September to October 2021 (Hasell et al., 2020).

**Commented [HK1]:** Do not include the figure of covid cases, since we do not have permission for use it

**Commented [LG2R1]:** Professor, at the right bottom it says CC BY. I could also remove if still desired.

Short description: <u>CC BY</u>: This license allows reusers to distribute, remix, adapt, and build upon the material in any medium or format, so long as attribution is given to the creator. The license allows for commercial use.

Similarly, we generated word clouds for the posts collected in the observed peak months of March 2020 and September 2021 (figure X and X).



Figure 2 - Word cloud - Rural, March 2020



Figure 3 - Word cloud - Rural, September 2021

While the results from the word clouds indicate a difference between the causes of the two peaks (predominantly linked to supermarkets in the first moment, and petrol in the second), the peaks are similar in the month of September 2021 between rural and urban posts. However, one can observe that, in the word cloud of rural areas, the words founds are relate to the Covid-19 event, whilst in urban areas ththe words are more associated with transport and energy.

# While this analysis is useful for assessing association, it is not able to assess the causality effect between rural-areas and fake-news or panic buying content spread. This is due to the fact that indirect unmeasured causes may lead to these differences, such as those that characterize rural and urban areas in the first place. In other words, what defines rural (or urban) areas is never a single characteristic, but multiple factors that co-influence the outcome.

However, when looking at the statistical association, total collected posts present a correlation of 98% when looking only at the second half of the data (from January 2021). If we look only at the first half (from March 2020), the correlation drops to 88%. Although correlations are relatively high, this can be due to the fact that many months present little to no posts on both cases. The quantitative and qualitative results seem to present a direct relationship with key events in the UK, evidencing possible cause and effect, here measured between (unknown origin) cause and posts, as further discussed in the next section.

# 5. Discussion

Findings revealed that fake news posts constitute around one third of panic buying ones showing a significant existence of rumors on social media associated with panic buying. This mandates the need to set measures to reduce fake news propagation. Moreover, comparing rural and urban activities on Twitter highlights primarily that there are more posts on panic buying as well as more posts on fake news in rural areas. Remarkably, in March 2020, high posts on panic buying were observed (but relatively lower percentage of fake news than in September 2021) unlike urban areas where posts barely include panic buying or fake news posts. Throughout the duration from March 2020 to September 2021, posts on both subjects are remarkably less (or sometimes were not existing) in urban areas compared to rural ones.

# -- - .

The outcome of this study illustrates that rural areas are more engaged in social media and users in these locations exchange information about panic buying and fake news. The study also shows a significantly higher volume of fake news posts at the onset of the COVID-19 pandemic in March 2020 as shown in figure (x) in rural areas compared to urban areas. The findings also show a significant association of these fake news to panic buying posts on Twitter. Moreover, we can see that the word clouds in rural areas in March 2020 show further evidence on this panic buying behaviour as reflected on the keywords appearing on tweets and related to panic buying such as 'toilet rolls', 'Coronavirus'; 'food', 'shop' along with mentioning of many super market chains such as 'Sainsbury's' and 'Asda'. Whilst in urban areas posts classified as fake news or those indicating a panic buying behaviour were minimal. Here, when Bentall et al. (2021) examined the buying behavior of consumers during the pandemic, they found that over-purchasing is higher in urban than rural areas. We argue that these results show the impact of the digital divide on fake news circulation and panic buying. As we noted earlier, fewer tech savvy users in rural areas are potentially less likely to question fake news and hence engage in more panic buying. As such, this could be due to the gap in ICT readiness that exists between urban and rural areas (Asfar and Zainuddin, 2015; Gilliggan, 2005). Being at an early stage of readiness when it comes to the use of the Internet, could make users in rural areas less experienced in scrutinizing overstated statements or fake news before sharing them. However, we must be cautious in interpreting these results taking into consideration that in rural areas and due to their location, they can understandably be more fearful of product unavailability as well as of contracting COVID-19 where access to healthcare at times of a pandemic could be more challenging (Whelan et. Al. 2021). And also taking into consideration that social media content does not necessarily or fully reflect the off-line reality. Moreover, whether panic buying behavior is high on social media or physically in rural areas, in both situations, it is vital to understand the context and culture of social media users living in these locations to have an in-depth insight about their attitudes and activities toward panic buying.

Gilliggan (2005) highlighted interesting insights about the lifestyle of people living in rural areas and its effect on their adoption and use of technology. He noted that people living in rural areas experience scarcity in a number of resources such as, information, goods, and services than cities' residents. Large retail stores, being usually far from their homes in addition to less developed network of transportation, makes it hard for them to obtain all the products or services they need. Mobility restrictions limit them also from obtaining adequate health services, meeting with others out of their physical circle, or even having access to plenty of leisure options. Consequently, they could rely more on mobile devices as a mean for maintaining their relationships with others and for getting a wider exposure and better access to resources. This could explain the fact that the number of mobile subscribers in rural area is higher than urban ones in several countries including the UK (Oftel, 2002). The context in which consumers exist was therefore considered in a number of studies that investigated panic buying. As an example, through applying the Stimulus-Organism-Response model (SOR), Islam et al. (2020) and Li et al. (2021) confirmed that the environment is pivotal as a stimulus that can affect consumers' feeling towards impulsive buying.

In the same vein, our analysis in September 2021, shows consistent results where engagement with fake news and panic buying in rural areas is significantly higher than in urban areas, yet, on this occasion there was a noticeable increase in urban areas behaviour compared to March 2020. This could be due to more employees returning to work from office instead of working from home. Hence, the lack of petrol could have a huge impact on their daily lives. Moreover, this is in line with the extant literature where "The sudden increase in buying of essentials creates imbalance between supply and demand, and this exerts more pressure in rural areas and people with low income (Arafat et al., 2020)" and once again

means that rural areas feel more pressure and fearful of the impact of lack of a necessary commodity like fuel on their daily lives.

This paper contributes to the extant literature on consumer involvement on social networking sites. More specifically, the study offers novel findings on the intersectionality between fake news on social media platforms, panic buying and digital divide. Interestingly, the paper shows that whilst significant efforts and resources are devoted to address the digital divide in the UK, more effort is needed to address the ICT readiness in rural areas. As not doing so with enough speed/ effectiveness could lead (as our findings show) to users in rural areas being more prone to the adverse effects of being connected on social media platforms. This contribution is related to a very peculiar circumstances i.e. covid-19 and hence we acknowledge that this might have exacerbated the negatives. Still, this underlines the importance for more attention to addressing the ICT readiness in rural areas especially with a greater reliance on the virtual world as we move forward.

One of the contribution of our paper lies in distinguishing the differences and similarities between both rural and urban inhabitants with regard to the interrelations between the above three concepts, as most studies concentrate on urban areas (Gilliggan, 2005). In this context, our paper represents one step in studying the impact of fake news generated and spread over social media platforms on panic buying, comparing results from different geographical areas, which can imply distinct behaviors, uses, and experience with social media. Further studies are planned to develop a framework that encompasses other dimensions that could influence consumers' over-purchasing patterns, and how these dimensions could be understood differently based on different consumers' contexts and personal traits.

Overall our study has highlighted important subtle dynamics related to the concepts of fake news, panic buying, and digital divide. Here, the study has shown that more panic buying and fake news were observed in rural areas, with a more severe impact at the onset of the Covid-19 pandemic. The same behavior was not observed in urban areas. This result suggest that the digital divide and other socio-cultural constructs may have relevant roles in panic buying. Some constructs may include for example digital skills, users' level of awareness, economic conditions, supply of products and services in rural or urban areas, and the ability to scrutinise and moderate the information on social media. The severity of panic buying and fake news were somehow lower in September 2021 compared to March 2020. Therefore, there is relevant time dimension related to the nature of the event and the perceived implications or associated risks.

Rural dwellers must have perceived much higher risk associated with getting Covid-19 at a time when the world knew nothing about the virus in comparison with a time when there would already be a vaccine. The existence of a vaccine could have mitigated the perceived associated risks, Understandably, these associated risks are lower when the issue in hand is related to an energy crisis (despite its significant impact of daily life). Similarly, and in terms of the volume of fake news, the digital divide could be observed in terms of a higher level of fake news consumption in rural areas

**Commented [HK3]:** Please see whether these arguments are adequate. In the conclusion, these questions have to be revisited

during the onset of the pandemic in March 2020. Our study has also documented a higher level of panic buying in rural areas during the sampled periods, which goes against the extant literature. This highlights the impact of the digital divide on fake news circulation and panic buying. Here, the study emphasises the importance of understanding the socio-economic and demographic realities in relation to the ongoing digital divide, as these seems to affect SNSs users' behaviour.

The study offers an important theoretical contribution regarding a subtle interplay between the concepts of digital divide, panic buying and fake news within the context of social media and shows a correlation between these constructs.

From a practical perspective, the study highlights that while social networking sites (SNSs) offer many opportunities for equality and inclusion, it opens the doors for unintended consequences that varies subject to the socio-economic context in question. Hence, more effort is needed in order to enhance social media governance where more safeguards are enabled to spot and spot/stop circulation of fake news. This can happen by further collaboration between SNSs companies and government agencies and relevant charitable and NGOs to identify features and characteristics of urban v. rural communities. From a managerial perspective, SNSs companies could then introduce these advanced features to protect users and minimise circulation of fake news.

The study has also important policy implications, as it highlights that addressing the fake news and any negative consequences of engaging with SNSs, especially in more deprived areas, this can only be achieved through better social media governance. Hence, policy makers need to bring various stakeholders including social media users, social media operators, legislators, non-governmental organisations to have a conscious dialogue on how to improve SNSs governance without affecting freedom of speech and other rights enjoyed in democratic societies. Indeed, careful consideration for digital divide and related skills gap should be a top priority.

## 6. Conclusions, Limitations, and Future Research

Our analysis demonstrates that despite a significant effort is being made to improve digital connectivity of rural areas in the UK, at time of crises, this connectivity could mean being more prone to experience adverse effects of being connected i.e. inability to spot fake news and engagement with a panic buying behaviour. These adverse impacts may require more effective training strategies and digital awareness, particularly in relation to social media platforms which constitutes an integral and crucial part of our daily lives. It is equally important to raise awareness about the shortcomings and even dangers of engaging with social platforms.

As Covid-19 is a global phenomenon that has a significant socio-economic effect, it opens a venue for plenty of studies aiming at understanding how social change in general and consumers' behavior in particular can be shaped by technology. Consumer buying behavior is a complicated and multidimensional concept especially in the existence of social commerce

#### Commented [HK4]: did not understand the argument

**Commented [ks5R4]:** I have made some amendmentshope it is clearer now. Please note that all text in red was addressing reviewer's two comments. Thank you (Aragoncillo and Orus, 2018; Abdelsalam et al., 2020; Naeem, 2021); mandating further studies in this field of research.

Our paper seeks to provide a contribution in this direction. While there is a body of literature that investigated the effect of social media on panic buying, there is scarcity of studies in examining their association with fake news. The only study that combined the three concepts related to fake news, panic buying, and social media was from Naeem and Ozuem (2021b) that tried to understand how fake news over social media platforms affect consumers' emotions and rationalities driving them to panic buying intentions and actions.

One possible limitation for the research is the data collection step. One reason for this relies on the fact that we depend both on what we input for the twitter's search engine Application Programming Interface (the actual list of towns and counties being used is not extensively elaborated for the UK) as well as on the inner workings of the search engine itself, which could theoretically handle posts from urban and rural areas differently, hence resulting in a difference in proportion between urban and rural areas relative to the actual total number of posts generated on the platform. One general solution for this issue could be the development on future endeavours of a broader collection process, both including more areas of search for twitter as well as the collection from other sources, such as local news outlets.

A second limitation present on this work is the lack of available good and robust databases for training fake news classifiers. The current databases available on this subject for research are either reference specific and do not generalize well for the identification of universal fake news or do not have an adequate volume for training and generalization. The one used on this work seems to have a sufficient balance in quality and total volume, however it's outdated examples could reflect in a decrease in accuracy. The research could profit from future investigations of the real accuracy of the developed classification model, most likely through a manual classification and comparison of a sufficient sample as well as the collection and structuring of an updated version of the LIAR Dataset, containing e.g. COVID-19 content.

The propagation of fake news can lead to threats of panic buying of groceries, which consequently increases the anxiety of vulnerable people leading to depression and anxiety that provokes further panic buying practices (Naeem and Ozuem, 2021b). Moreover, rumors and misinformation could cause supply chain disruption (Loxton et al., 2020). Fake news also

poses a serious challenge to businesses, for maintaining their operations under the COVID-19 restrictions enforced by the authorities. Conflicting information concerning the pandemic's prevention and treatment, and unclear statements regarding government regulations encourage people to stockpile leading to price rise and to scams' occurrence (Dang, 2021). Naeem (2021) pointed out to the ambiguities caused by the late or ineffective response of the government to people's concerns and to the weak communication and integration between governmental and non-governmental institutions, which reduce the trust of citizens towards their governments (Naeem and Ozuem, 2021b) opening rooms for influencers and non-professionals to post false or inaccurate information (Din et al., 2020) indulging them in additional panic buying behavior. Therefore, prompt responses, better institutional cooperation and unified communication strategies are vital to address panic buying during crises. Furthermore, it is recommended that governments and civil societies collaborate in shaping the required policies to address the negative consequences of fake news. Din et al. (2020) stressed on the importance of education and knowledge of societies that enable individuals to possess a critically thinking of the credibility of the flooding online content they encounter. Rather than being a source for disseminating fake news, social media could realize remarkable benefits through providing people with valuable and timely information (Donovan, 2020). Arafat, et al., (2020) have claimed that the identification of fake news and ignoring them or reporting about it to appropriate authority may also be beneficial in the prevention of panic buying.

# 7. REFERENCES

Abdelsalam, S., Salim, N., Alias, R.A., and Husain, O. (2020). "Understanding Online Impulse Buying Behavior in Social Commerce: A Systematic Literature Review. IEEE Access 8, 89041– 89058.

Akamatsu, N., and Mizuno, M. (2021). "Consumer Panic Buying: Understanding the Behavioral and Psychological Aspects. Online 19/12/2021 at http://dx.doi.org/10.2139/ssrn.3796825

Alalwan, A. A. (2018). "Investigating the Impact of Social Media Advertising Features on Customer Purchase Intention". International Journal of Information Management 42: 65–77.

Alam, F., Shaar, S., Dalvi, F., Sajjad, H., Nikolov, A., Mubarak, H., Martino, G.D.S., Abdelali, A., Durrani, N., Darwish, K. and Al-Homaid, A., 2020. Fighting the COVID-19 infodemic: modeling the perspective of journalists, fact-checkers, social media platforms, policy makers, and the society. arXiv preprint arXiv:2005.00033.

Anderson, R.M., Heesterbeek, H., Klinkenberg, D. and Hollingsworth, T.D. (2020). "How will Country-Based Mitigation Measures Influence the Course of the COVID-19 Epidemic?". The Lancet 395(10228): pp. 931-934.

Appel, G., Grewal, L., Hadi, R., & Stephen, A. T. (2020). "The Future of Social Media in Marketing. Journal of the Academy of Marketing Science, 48(1): 79–95.

Aragoncillo, L. and Orus, C. (2018). "Impulse Buying Behaviour: An Online-Offline Comparative and the Impact of social media. Spanish Journal of Marketing - ESIC 22 (1), 42–62. https://doi.org/10.1108/SJME-03-2018-007.

Arora, Raj (1982). "Validation of an S-O-R Model for Situation, Enduring, and Response Components of Involvement". Journal of Marketing Research, 19 (Nov), 505-516.

Asfar, N. and Zainuddin, Z. (2015). "Secondary Students' perceptions of Information, Communication and Technology (ICT) Use in Promoting self-Directed Learning in Malaysia. The Online Journal of Distance Education and e-Learning 3(4): 67-82.

Asur, S. and Huberman, B.A., 2010, August. Predicting the future with social media. In 2010 IEEE/WIC/ACM international conference on web intelligence and intelligent agent technology (Vol. 1, pp. 492-499). IEEE.

Baker Qureshi, P.A., Murtaza, F., and Kazi, A.G. (2019). "The Impact of Social Media on Impulse Buying Behaviour in Hyderabad Sindh Pakistan". International Journal of Entrepreneurial Research 2(2): 8–12.

Ballantine, P. W. (2013). "Changes in Retail Shopping Behaviour in the Aftermath of an Earthquake". The International Review of Retail, Distribution and Consumer Research 24(1): 28–42.

Barnes, S., Diaz, M., and Arnaboldi, M. (2020). "Understanding Panic Buying during COVID-19: A Text Analytics Approach". Expert Systems With Applications. Online 8/12/2021 at https://doi.org/10.1016/j.eswa.2020.114360

Barr, S. (2020). "Coronavirus Banic-buying: As Supermarkets Ration Items, Should Customers be Stockpiling?" Independent. Online 12/10/2021 at https://www.independent.co.uk/lifestyle/ food-and-drink/coronavirus-stockpile-emergency-list-food-hand-sanitiser-panic-b uying-a9373061.html

Bentall, R.P., Lloyd, A., Bennett, K., McKay, R., Mason, L., Murphy, J., et al. (2021). "Pandemic Buying: Testing a Psychological Model of Over-Purchasing and Panic Buying using Data from the United Kingdom and the Republic of Ireland during the Early Phase of the COVID-19 Pandemic. PLoS ONE 16(1): e0246339. Online 14/12/2021 at https://doi.org/10.1371/journal.pone.0246339

Besson, E.K. (2020). "COVID-19 (Coronavirus): Panic Buying and Its Impact on Global Health Supply Chains". World Bank. Online 12/10/2021 at https://blogs.worldbank.org/health/covid-19-coronavirus-panic-buying-and-its-impactglobal-health-supply-chains

Bhatti, Z., Arain, G., Akram, M. Fang, Y., and Yasina, H. (2020). "Constructive Voice Behavior for Social Change on Social Networking Sites: A Reflection of Moral Identity". Technological Forecasting & Social Change 157(2020), 120101.

Billore, S., Anisimova, T. (2021). Panic buying research: A systematic literature review and future research agenda. International Journal of Consumer Studies, 45(4), 777-804.

Blank, G., 2017. The digital divide among Twitter users and its implications for social research. Social Science Computer Review, 35(6), pp.679-697.

Blank, G. and Dutton, W.H., 2013. The Emergence of Next-Generation Internet Users. A companion to new media dynamics, pp.122-141.

Blank, G., and Groselj, D. (2015). Examining Internet use through a Weberian lens. International Journal of Communication.

Campbell, M. C., Inman, J. J., Kirmani, A., Price, L. L. (2020). In Times of Trouble: A Framework for Understanding Consumers' Responses to Threats. Journal of Consumer Research, 47(3), 311–326.

Castellano, S., & Dutot, V. (2017). Investigating the influence of E-word-of-mouth on E-reputation. International Studies of Management & Organization, 47(1), 42-60.

Chamberlain, J., Turpin, B., Ali, M., Chatsiou, K. and O'Callaghan, K., 2021. Designing for collective intelligence and community resilience on social networks. Human Computation, 8(2), pp.15-32.

Cinelli, M., Quattrociocchi, W., Galeazzi, A., Valensise, C.M., Brugnoli, E., Schmidt, A.L., Zola, P., Zollo, F. and Scala, A., 2020. The COVID-19 social media infodemic. Scientific Reports, 10(1), pp.1-10.

Cogley, M. (2020). "Has Social Media Turbocharged Panic Buying by UK Shoppers? Online 12/10/2021 at https://www.telegraph.co.uk/technology/2020/03/10/has-social-media-turbocharged-panic-buying-uk-shoppers/

Collins, H.M. (1983). "An Empirical Relativist Programme in the Sociology of Scientific Knowledge". In: Knorr-Cetina, K., Mulkay, M. (Eds.), Science Observed: Perspectives on the Social Study of Science, London: Sage, pp. 85–114.

Collinson, P. (2020). "Panic Buying on Wane as Online Shopping Takes Over, Bays bank". TheGuardian.Online12/10/2021athttps://www.theguardian.com/business/2020/mar/30/coronavirus-bank-finds-end-to-panic-buying-while-online-shopping-takes-over

Cruz-Cárdenas, J., Zabelina, E., Guadalupe-Lanas, J.,Palacio-Fierro, A., Ramos-Galarza, C. (2021). COVID-19, consumer behavior, technology, and society: A literature review and bibliometric analysis, Technological Forecasting and Social Change, 173, 121179. Dang, H. L. (2021). "Social Media, Fake News, and the COVID-19 Pandemic: Sketching the Case of Southeast Asia". Austrian Journal of South-East Asian Studies, 14(1): 37-57.

Department for Digital, C., Media & Sport. (2021). Building Digital UK. Retrieved from https://www.gov.uk/guidance/building-digital-uk#public-reviews

Depoux, A., Martin, S., Karafillakis, E., Preet, R., Wilder-Smith, A. and Larson, H., 2020. The pandemic of social media panic travels faster than the COVID-19 outbreak. Journal of travel medicine, 27(3), p.taaa031.

DiMaggio, P., Hargittai, E., Neuman, W. R., & Robinson, J. P. (2001). Social implications of the Internet. Annual review of sociology, 27(1), 307-336.

Din, M., Kamal, N. and Abdullah, A. (2020). "Fake News on Social Media during COVID-19 Crisis". Proceedings of the Philosophy of Science and Civilization Seminar 2 (2) 2020. Online 5 November at https://www.researchgate.net/publication/342747252\_Prosiding\_Seminar\_Falafah\_Sains\_d an\_Ketamadunan\_22\_2020#page=26

Donovan, J. (2020). "Here's how Social Media can Combat the Coronavirus 'Infodemic'". MITTechnologyReview.Online30Decemberathttps://www.technologyreview.com/2020/03/17/905279/facebook-twitter-social-media-infodemic-misinformation/

Dulam,R., Furuta.,K., and Kanno, T. (2021). "Consumer Panic Buying: Realizing Its Consequences and Repercussions on the Supply Chain". Sustainability 2021,13,4370. Online 14/12/2021 at https:// doi.org/10.3390/su13084370

Dutton, W.H. and Blank, G., 2011. Next generation users: the internet in Britain.

Fishbein, M. & Ajzen, I. (2011). "Predicting and Changing Behavior: The Reasoned Action Approach". New York. <u>https://doi.org/10.4324/9780203838020</u>

Forbes, S.L. (2017), "Post-Disaster Consumption: Analysis from the 2011 Christchurch Earthquake", The International Review of Retail, Distribution and Consumer Research 27(1): pp. 28-42.

Frick, T., Tsekauras, D. and Li, T., 2014. The times they are a-changin: Examining the impact of social media on music album sales and piracy. In Annual Meeting of the Academy of Management.

Gallagher, C. E., Watt, M. C., Weaver, A. D., and Murphy, K. A. (2017). "I fear, therefore, I Shop! Exploring Anxiety Sensitivity in Relation to Compulsive Buying". Personality and Individual Differences, 104, pp:37–42.

Gerli, Paolo, and Jason Whalley. "Fibre to the countryside: A comparison of public and community initiatives tackling the rural digital divide in the UK." Telecommunications Policy 45.10 (2021): 102222.

Gilligan, R. (2005). "Questioning the 'Rural' Adoption and Use of ICTs". in Haddon, L., Mante-Meijer, E., Sapio, B., Kommonen, K-H, Fortunati, L., and Kant, A (eds) Everyday Innovators, Researching the Role of Users in Shaping ICTs, Springer, Dordrect, 155-167.

Gruhl, D., Guha, R., Kumar, R., Novak, J. and Tomkins, A., 2005, August. The predictive power of online chatter. In Proceedings of the eleventh ACM SIGKDD international conference on Knowledge discovery in data mining (pp. 78-87).

Habersaat, K.B., Betsch, C., Danchin, M., Sunstein, C.R., Böhm, R., Falk, A., Brewer, N.T., Omer, S.B., Scherzer, M., Sah, S. and Fischer, E.F., 2020. Ten considerations for effectively managing the COVID-19 transition. Nature human behaviour, 4(7), pp.677-687.

Hall, C.M., Prayag, G., Fieger, P., and Dyason, D. (2020). "Beyond Panic Buying: Consumption Displacement and COVID-19". Journal of Service Management 32(1): pp. 113-128.

Hall, C.M., Scott, D., and G€ossling, S. (2020), "Pandemics, transformations and tourism: be careful what you wish for", Tourism Geographies 22(3): pp. 577-598.

Hennig-Thurau, T., Wiertz, C. and Feldhaus, F., 2015. Does Twitter matter? The impact of microblogging word of mouth on consumers' adoption of new movies. Journal of the Academy of Marketing Science, 43(3), pp.375-394.

Hine, C. (2020). The evolution and diversification of Twitter as a cultural artefact in the British press 2007–2014. Journalism Studies, 21(5), 678-696.

Huberty, M. (2015). Can we vote with our tweet? On the perennial difficulty of election forecasting with social media. International Journal of Forecasting, 31(3), 992-1007.

Hasell, J., Mathieu, E., Beltekian, D. et al. A cross-country database of COVID-19 testing. Sci Data 7, 345 (2020). https://doi.org/10.1038/s41597-020-00688-8

Hou, Z., Du, F., Zhou, X., Jiang, H., Martin, S., Larson, H., and Lin, L. (2020). "Cross-Country Comparison of Public Awareness, Rumors, and Behavioral Responses to the COVID-19 Epidemic: Infodemiology Study". Journal of Medical Internet Research, 22(8), e21143.

Huan, C., Park, S., Kang, J. (2021). Panic Buying: Modeling What Drives it and How it Deteriorates Emotional Well-being. Family and Consumer Sciences Research Journal, 50(2), 150-164.

Islam, T., Pitafi, A.H., Arya, V., Wang, Y., Akhtar, N., Mubarik, S., and Xiaobei, L. (2020). "Panic

buying in the COVID-19 pandemic: A multi-country examination". Journal of Retailing and Consumer Services 59, 102357.

Kadam, A. B., & Atre, S. R. (2020). Negative impact of social media panic during the COVID-19 outbreak in India. Journal of travel medicine, 27(3), taaa057.

Kim, S.-B., Han, K.-S., Rim, H.-C., Myaeng, S.H., 2006. Some effective techniques for naive bayes text classification. IEEE transactions on knowledge and data engineering 18, 1457–1466.

Kulemeka, O. (2010). "US Consumers and Disaster: Observing "Panic Buying" during the Winter Storm and Hurricane Seasons". Advances in Consumer Research 37: 837–38.

Laato, S., Islam, N., Farooq, A., and Dhir, A. (2020). "Unusual purchasing behavior during the early stages of the COVID-19 pandemic: the stimulus-organism-response approach". Journal of Retailing and Consumer Services 57, 102224

Lee, H., Padmanabhan, V. and Whangm, S. (1997). "Information Distortion in a Supply Chain: The Bullwhip Effect". Management Science, 43(4), pp: 546–558.

Lehberger, M., Kleih, A.K., and Sparke, K. (2021). "Panic buying in Times of Coronavirus (COVID19): Extending The Theory of Planned Behavior to Understand the Stockpiling of Nonperishable Food in Germany. Appetite, 161, Online 20/9/2022 at https://doi.org/10.1016/j.appet.2021.105118.

Li, X., Zhou, Y., Wong, Y.D., Wang, X., and Yuen, K.F. (2021). "What Influences Panic Buying behaviour? A Model Based on Dual-System Theory and Stimulus-Organism-Response Framework". International Journal of Disaster Risk Reduction 64 (October 2021), 102484 Lopez, C.E., Vasu, M. and Gallemore, C., 2020. Understanding the perception of COVID-19 policies by mining a multilanguage Twitter dataset. arXiv preprint arXiv:2003.10359. Loxton, M., Truskett, R., Scarf, B., Sindone, L., Baldry, G., and Zhao, Y. (2020). "Consumer Behaviour during Crises: Preliminary Research on How Coronavirus Has Manifested Consumer Panic Buying, Herd Mentality, Changing Discretionary Spending and the Role of the Media in Influencing Behaviour". Journal of Risk and Financial Management 13(166): pp.1-21.

McCallum, A., Nigam, K., 1998. A comparison of event models for naive bayes text classification. Presented at the AAAI-98 workshop on learning for text categorization, Citeseer, pp. 41–48.

Mao, F. (2020). "Coronavirus Panic: Why are People Stockpiling Toilet Paper? Online 12/10/2021 at https://www.bbc.co.uk/news/world-australia-51731422

Muqadas, F., Rehman, M., Aslam, U., and Ur-Rahman, U. (2017). "Exploring the Challenges, Trends and Issues for Knowledge Sharing: A Study on Employees in Public Sector Universities". VINE Journal of Information and Knowledge Management Systems 47(1): 2–15.

Naeem, M. and Ozuem, Z. (2021a). "Customers' social interactions and panic buying behavior: Insights from social media practices". Journal of Consumer Behaviour 20(5): 1191-1203.

Naeem, M. and Ozuem, Z. (2021b). "Understanding Misinformation and Rumors that generated panic Buying as Social Practice during COVID-19 Pandemic: Evidence from Twitter, YouTube, and Focus Group Interviews". Information Technology and People. Online 20 December at https://www.emerald.com/insight/content/doi/10.1108/ITP-01-2021-0061/full/pdf?title=understanding-misinformation-and-rumors-that-generated-panic-buying-as-a-social-practice-during-covid-19-pandemic-evidence-from-twitter-youtube-and-focus-group-interviews

Naeem, M. (2021). "Do Social Media Platforms Develop Consumer Panic Buying during theFear of Covid-19 pandemic". Journal of Retailing and Consumer Services, 58 (2021) 102226.Online20Novemberat

https://www.sciencedirect.com/science/article/pii/S0969698920309814Nakano, S., Oftel (Office of Telecommunications). Consumers' use of mobile telephony; 2002. Oftel, London. Ossewaarde, M., 2019. Digital transformation and the renewal of social theory: Unpacking the new fraudulent myths and misplaced metaphors. Technological Forecasting and Social Change, 146, pp.24-30.

Peck, H. (2006). "Resilience in the Food Chain: A Study of Business Continuity Management in the Food and Drink Industry". Cranfield University, Shrivenham.

Pennycook, G., McPhetres, J., Zhang, Y. and Rand, D., 2020. Fighting COVID-19 misinformation on social media: Experimental evidence for a scalable accuracy nudge intervention. PsyArXiv Preprints, 10.

Philip, L., Cottrill, C., Farrington, J., Williams, F., & Ashmore, F. (2017). The digital divide: Patterns, policy and scenarios for connecting the 'final few'in rural communities across Great Britain. Journal of Rural Studies, 54, 386-398.

Prentice, C., Quach, S., Thaichon, P. (2022). Antecedents and consequences of panic buying: The case of COVID-19. International Journal of Consumer Studies, 46(1), 132-146.

Pulido, C.M., Villarejo-Carballido, B., Redondo-Sama, G. and Gómez, A., 2020. COVID-19 infodemic: More retweets for science-based information on coronavirus than for false information. International Sociology, 35(4), pp.377-392.

Quarantelli, E.L. (1999). "Disaster Related Social Behavior: Summary of 50 Years of Research Findings". Disaster Research Center. Online 7/12/2021 at https://udspace.udel.edu/bitstream/handle/19716/289/PP%20280.pdf?sequence=1&isAllo wed=y

Roggeveen, A.L. and Johar, G.V., 2002. The Role of Evoked Range in the Integration of Discrepant Sales Forecasts: Process and Resultant Bias. ACR Asia-Pacific Advances.

Rook, W., and Fisher, R. J. (1995). "Normative Influence on Impulsive Buying Behavior". Journal of Consumer Research, 22(December), pp: 305-313.

Roser, M., Ritchie, H., Ortiz-Ospina, E. and Hasell, J., 2020. Coronavirus pandemic (COVID-19). Our world in data. Sheu, J.B. and Kuo, H.T. (2020), "Dual Speculative Hoarding: A Wholesaler-Retailer Channel Behavioral Phenomenon behind Potential Natural Hazard Threats". International Journal of Disaster Risk Reduction 44, Art.101430.

Salemink, K., Strijker, D., & Bosworth, G. (2017). Rural development in the digital age: A systematic literature review on unequal ICT availability, adoption, and use in rural areas. Journal of Rural Studies, 54, 360-371.

Serafino, P. (2019). Exploring the UK's digital divide. Office for National Statistics.

Sim, K., Chua, H.C., Vieta, E., and Fernandez, G. (2020). "The Anatomy of Panic Buying Related to the Current COVID-19 Pandemic". Psychiatry Research. Online 12/12/2021 at https://www.ncbi.nlm.nih.gov/pmc/articles/PMC7158779/

Singh, G., Aiyub, A.S., Greig, T., Naidu, S., Sewak, A. and Sharma, S. (2021), "Exploring panic buying behavior during the COVID-19 pandemic: a developing country perspective", International Journal of Emerging Markets, Vol. ahead-of-print No. ahead-of-print. https://doi.org/10.1108/IJOEM-03-2021-0308

Slovic, P., Finucane, M. L., Peters, E., and Macgregor, D. G. (2004). "Risk as Analysis and Risk as Feelings: Some Thoughts about Affect, Reason, Risk, and Rationality". Risk Analysis 24(2): 311–22.

Sneath, J. Z., Lacey, R. and Kennett-Hensel, P. A. (2009). "Coping with a Natural Disaster: Losses, Emotions, and Impulsive and Compulsive Buying". Marketing Letters, 20(1), pp: 45– 60.

Sparks, C. (2013). What is the "Digital Divide" and why is it important? Javnost-The Public, 20(2), 27-46.

Sterman, J. D., and Dogan, G. (2015). "I'm not Hoarding, I'm just Stocking up before the Hoarders Get Here: Behavioural Causes of Phantom Ordering in Supply Chains". Journal of Operations Management, 39, pp: 6–22.

Tan, K., Sia, J., ND TANG, d. (2021). "To Verify or Not to Verify: Using Partial Least Squares to Predict Effect of Online News on Panic Buying during Pandemic". Asia Pacific Journal of Marketing and Logistics, 34(4), pp:647-668.

Tasnim, S., Hossain, M.M. and Mazumder, H., 2020. Impact of rumors and misinformation on COVID-19 in social media. Journal of preventive medicine and public health, 53(3), pp.171-174.

Thomas, J.A. (2014). "Community Resilience, Latent Resources and Resource Scarcity after an Earthquake: Is Society really Three meals away from Anarchy?". Natural Hazards 74(2): pp. 477–490.

Tsao, Y.C., Raj, P.V.R.P., and Yu, V. (2019). "Product Substitution in Different Weights and Brands Considering Customer Segmentation and Panic Buying Behavior. Industrial Marketing Management 77: 209–220.

Van Bavel, J. J., Baicker, K., Boggio, P. S., Capraro, V., Cichocka, A., Cikara, M., et al. (2020). Using social and behavioural science to support COVID-19 pandemic response. Nat. Hum. Behav. 4, 460–471. doi: 10.1038/s41562-020-0884-z

van Der Linden, S., Roozenbeek, J. and Compton, J., 2020. Inoculating against fake news about COVID-19. Frontiers in psychology, 11, p.2928.

World Health Organization (2020). Novel Coronavirus (2019-nCoV) Situation Report - 13. Available online at: https://www.who.int/docs/default-source/ coronaviruse/situation-reports/20200202-sitrep-13-ncov-v3.pdf (accessed February 2, 2020).

World Health Organization (2022). WHO Coronavirus (COVID019) dashboard. https://covid19.who.int/ (accessed October 6th, 2022).

Yoon, J., Ram Narasimhan, M., and Kim, M.K. (2017). "Retailer's Sourcing Strategy under Consumer Stockpiling in Anticipation of Supply Disruptions". International Journal of Production Research 56(10): 3615–35.

Yuen, K.F., Wang, X., Ma, F., and Li, K.X. (2020). "The Psychological Causes of Panic Buying Following a Health Crisis". International Journal of Environmental Research and Public Health 17(10): 3513.

Wang, W.Y., 2017. "liar, liar pants on fire": A new benchmark dataset for fake news detection. arXiv preprint arXiv:1705.00648.

Watts, G. (2020). COVID-19 and the digital divide in the UK. The Lancet Digital Health, 2(8), e395-e396.

Zhang, H., 2004. The optimality of naive Bayes. AA 1, 3.

Zhang, Y., Jin, R., Zhou, Z.-H., 2010. Understanding bag-of-words model: a statistical framework. International Journal of Machine Learning and Cybernetics 1, 43–52.

Zheng, R., Shou, B., and Yang, J. (2020). "Supply Disruption Management under Consumer Panic Buying and Social Learning Effects, Omega. Online 14/12/2021 at https://doi.org/10.1016/j.omega.2020.102238

Zarocostas, J., 2021. How to fight an infodemic. Lancet 2020; 395: 676. J ALLERGY CLIN IMMUNOL PRACT JULY.